Author’s response to reviews

Title: Possible value of antifibrotic drugs in patients with progressive fibrosing non-IPF interstitial lung diseases

Authors:
Sebastiano Emanuele Torrisi (torrisiseby@hotmail.it)
Nicolas Kahn (Nicolas.Kahn@med.uni-heidelberg.de)
Julia Wälscher (Julia.Waelscher@rlk.uk-essen.de)
Nilab Sarmand (nilab.sarmand@med.uni-heidelberg.de)
Markus Polke (markus.polke@med.uni-heidelberg.de)
Kehler Lars (lars.kehrler@med.uni-heidelberg.de)
Monika Eichinger (monika.eichinger@med.uni-heidelberg.de)
Claus Peter Heussel (heussel@uni-heidelberg.de)
Stefano Palmucci (spalmucci@sirm.org)
Francesca Maria Sambataro (francesca.samba@gmail.com)
Gianluca Sambataro (dottorsambataro@gmail.com)
Domenico Sambataro (d.sambataro@hotmail.it)
Carlo Vancheri (vancheri@unict.it)
Michael Kreuter (kreuter@uni-heidelberg.de)

Version: 2 Date: 11 Sep 2019

Author’s response to reviews:

Thanks to the editor for his comments.
Editor Comments:

1. Authors’ Contributions

Please include a statement regarding the contributions of all authors under the “Authors’ Contribution” section of the Declarations. Please represent author’s names using their full initials, not their name. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

-> Thanks, a complete statement of authors contributions has been included in the appropriate section.

2. Table 2

Currently, the age and gender information present in Table 2, in combination with other identifiers, may compromise patient/participant anonymity.

If you do not have consent to publish, please amend the file to address this issue, by removing this information or by providing ages as age-ranges.

-> Thanks, Age and gender have been deleted from the table.

3. Figure 3

Please ensure that all figures, tables and supplementary/additional files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

-> Thanks again, Fig. 3 has been now correctly cited in the discussion paragraph.

4. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

-> a clean and updated version has been uploaded.